Author's response to reviews

Title: Intraluminal instillation of urokinase and autologous plasma: A method to unblock occluded central venous ports

Authors:
Georg Seifert (georg.seifert@charite.de)
Hanno Riess (hanno.riess@charite.de)
Karl H Seeger (karl.seeger@charite.de)
Guenter Henze (guenter.henze@charite.de)
Anja Borgmann (ania.borgmann@charite.de)

Version: 4 Date: 22 March 2006

Author's response to reviews:

Dear Editor,
please find attached our revised manuscript. We added the sentence "Informed consent was obtained from patients, as approved by the Institutional Ethical Committee" in Methods section. We revised the manuscript formatting and considered the suggestions of the referees.
We hope you may find our report now acceptable for publication.

Yours sincerely

Georg Seifert